Author's response to reviews

Title: Community mobilisation and health management committee strengthening to increase birth attendance by trained health workers in rural Makwanpur, Nepal: study protocol for a cluster randomised controlled trial

Authors:

Joanna Morrison (j.morrison@ich.ucl.ac.uk)
Kirti M Tumbahangphe (jetha_subba@hotmail.com)
Bharat Budhathoki (bmmagar3@gmail.com)
Rishi Neupane (r2062_baje@yahoo.com)
Aman Sen (amansen@gmail.com)
Kunta Dahal (kuntakd@yahoo.com)
Rita Thapa (kunsang_rita@yahoo.com)
Reema Manandhar (reema.sayami@gmail.com)
Dharma Manandhar (dsm@mira.org.np)
Anthony Costello (a.costello@ich.ucl.ac.uk)
David Osrin (d.osrin@ich.ucl.ac.uk)

Version: 4 Date: 13 May 2011

Author's response to reviews:

Dear Editors,

As requested we have made the following changes to our trial protocol:

We have changed the title to "Community mobilisation and health management committee strengthening to increase birth attendance by trained health workers in rural Makwanpur, Nepal: study protocol for a cluster randomised controlled trial"

We have included the full names institutions, addresses and email addresses, and indicated the corresponding author.

We have included a list of abbreviations and their meanings at the end of the manuscript before the reference list.

We have inserted a list of figures after the reference list.

We have removed the ethics and funding documents.

Many thanks for your consideration.

Joanna Morrison